\documentclass[11pt]{article}


\addtolength{\topmargin}{-1cm}
\addtolength{\textheight}{5cm}
\addtolength{\textwidth}{3cm}
\addtolength{\oddsidemargin}{-1.5cm}


\author{Teschl Gerald}
\title{Math 60}


\begin{document}

\thispagestyle{empty}


\begin{center}
\LARGE Welcome to \\
\LARGE Math 60: Finite Mathematics\\
\Large Fall 1995
\end{center} \vspace{1cm}

\noindent
\begin{tabular}{ll}
{\bf Instructor:} & Gerald TESCHL \\  \vspace*{1mm}
{\bf Office:} & 306 Mathematical Science Building \\  \vspace*{1mm}
{\bf Office Phone:} & 882-4267 \\  \vspace*{1mm}
{\bf Office Hours:} & MWF 3:40p.m.~-- 5:00p.m. or by appointment\\  \vspace*{1mm}
{\bf Textbook:} & Finite Mathematics 3th ed., Howard L. Rolf\\
{\bf Required materials:} & Math 60 	studentÿs handbook\\
& Four presealed exam packets\\
& TI 85 graphic calculator
\end{tabular}

\vspace{1cm}

\noindent
{\bf Grading:} \begin{minipage}[t]{11cm}
\begin{tabular}[t]{l@{\hspace*{3mm}}l}
Three hourly departmental exams	& 300 pts\\
Departmental final exam	& 200 pts\\
Homework and/or quizzes	& 100 pts\\
\hline
Total points possible	& 600 pts
\end{tabular} \end{minipage}

\vspace{1cm}

\noindent
{\bf Policies:}

	1.  Your {\bf homework} assignments will consist of:
\begin{itemize}
\item Previewing the material before it is presented in class and glancing over the 
corresponding homework problems. 
\item Reading and studying carefully the most recent material discussed in class and the
corresponding class notes;  doing the due homework problems.  
\end{itemize}
Homework may or may not be collected. Questions about homework problems will be
answered at the beginning of each class session.

	2. {\bf Attendance} will be taken daily starting the second week of classes. 
No points will be assigned for attendance, but any student with six or more absences will
be dropped from the roll.  


 3. {\bf Course Syllabus}.	You are required to purchase the Math 60 Studentÿs Handbook from
the Brady Bookstore. It includes the course syllabus, the schedule of exams, the homework
assignments, the homework supplements, the departmental policies which you should know
about, all supplemental instruction available to you free through the Learning Center,
test reviews, etc.

	4.  Weekly {\bf quizzes} will be given at 10 points each.  There will be no make-up
quizzes, but you will get to drop your two lowest quiz scores.  A quiz could be a homework
collected randomly! {\em If more than 5 students are absent a quiz is
assigned automatically.}

	5.  {\bf Make-up exams} will be given only in the event of a documented MU conflict,
a documented illness, or a death in the immediate family.  

	6.  {\bf Academic dishonesty} such as cheating and plagiarism is a very
serious offense and is grounds for disciplinary action - don't even think about it.

	7.  {\bf ADA}. If you have a special condition as addressed by the Americans with
Disabilities Act, let me know.


\end{document}
